Author's response to reviews

Title: Screening of exon 11 of BRCA1 gene using the High Resolution Melting approach for diagnostic in Moroccan breast cancer patients

Authors:

Meryam El Khachibi (el.khachibi.meryam@gmail.com)
Khalil Hamzi (khalil.hamzi@gmail.com)
Brehima Diakite (br.diakite@yahoo.fr)
Mohamed A Senhaji (m.amine.senhaji@gmail.com)
Amina Bakhchane (amina_bakhchane@yahoo.fr)
Amal Tazzite (amal_tazzite@yahoo.fr)
Hassan Jouhadi (hjouhadi@yahoo.fr)
Abdelhamid Barakat (hamid.barakat@pasteur.ma)
Abdellatif Benieder (benider.abdel@yahoo.fr)
Sellama Nadifi (nadifisel@yahoo.fr)

Version: 8
Date: 30 December 2014

Author's response to reviews:

To The Editor,

Sub: Re-Submission of Manuscript for publication

Dear Editor,

We would like to resubmit a revised version of the article “Screening of exon 11 for BRCA1 using the High Resolution melting for diagnostic in Moroccan breast cancer patients” (Manuscript ID 2054609999128887) for publication in the journal Research articles.

You will find enclosed a new version of the manuscript, which would now include the modifications that you suggested.

Please do not hesitate would you need further information.

Best Regards
Sellama Nadifi

E-mail address: nadifisel@yahoo.fr

Signature
Sellama Nadifi
1. Please can you clarify whether you have consent to publish the clinical details in table 2, from the participants in question. If you do, please include a statement clarifying you have consent to publish the details in table 2 from participants. If you do not, please anonymize table 2 by removing age, gender and details such as family history.

# We have the patients' consent for publishing their data.

2. Please format your references according to our guidelines:

# I adjusted the references according to the guidelines.

3. Copy-edit:

# I edited the paper and tried to improve the writing style.

4. Please include an author’s contributions guideline, please read our guidelines for more details:

# I added the authors’ contributions section to the manuscript.

5. Please include an acknowledgements section. Please read our guidelines for more details:

# I added an acknowledgements section to the manuscript.